Author’s response to reviews

Title: Cardiac Left Ventricular Myocardial Tissue Density, evaluated by Computed Tomography and Autopsy

Authors:

Alexandra Gheorghe (svejme@hotmail.com)
Andreas Fuchs (mck601@alumni.ku.dk)
Christina Jacobsen (Christina.jacobsen@sund.ku.dk)
Klaus Kofoed (Klaus.Kofoed@regionh.dk)
Rasmus Møgelvang (Rasmus.Moegelvang.01@regionh.dk)
Niels Lynnerup (Niels.lynterup@sund.ku.dk)
Jytte Banner (jytte.banner@sund.ku.dk)

Version: 3 Date: 25 Feb 2019

Author’s response to reviews:

Dear Dirk Krüger and reviewer Jonathan Lessick,

Thank you for the thorough revisions and comments through this process. I very much appreciate the time you have put in reviewing this manuscript and it has been a good learning process for me and for future study, so thank you for that.

You are right about the ranges and it is typing mistake made on my part. The ranges are corrected in the manuscript.

The ranges are -65.3;25.5.